Author's response to reviews

Title: Contraceptive use following spontaneous and induced abortion and its association with family planning services in primary health care: results from a Brazilian longitudinal study

Authors:

Ana LV Borges (alvilela@usp.br)
Funmilola OlaOlorum (fmolaolorun@gmail.com)
Elizabeth Fujimori (efujimor@usp.br)
Luiza AK Hoga (kikatuca@usp.br)
Amy Tsui (atsui@jhu.edu)

Version: 3
Date: 24 September 2015

Author's response to reviews: see over
September 24, 2015

Dear Dr Joao Paulo Souza
The Reproductive Health Editorial Team

Please find attached a revised manuscript of MS: 1958772142152534 - Contraceptive use following spontaneous and induced abortion and its association with family planning services in primary health care: results from a Brazilian longitudinal study.

We have provided a point-by-point response to the concerns. Your suggestions were appropriate, and therefore, entirely accepted.

Best regards,
Ana Luiza Vilela Borges
Contraceptive use following spontaneous and induced abortion and its association with family planning services in primary health care: results from a Brazilian longitudinal study

Ana Luiza Vilela Borges¹§, Funmilola Olaolorun², Elizabeth Fujimori¹, Luiza Akiko Komura Hoga³, Amy Ong Tsui⁴

¹ Department of Public Health Nursing, University of São Paulo School of Nursing, Brazil
² Department of Preventive Medicine and Primary Care, University of Ibadan, Nigeria
³ Department of Maternal, Child and Psychiatric Care, University of São Paulo School of Nursing, Brazil
⁴ Department of Population, Family and Reproductive Health, Johns Hopkins Bloomberg School of Public Health, USA

Editor's review

We would like to request you to provide us with a new Table 2 with crude numbers and crude OR in addition to the existing AdjOR. If crude OR deviate substantially from the adjusted OR, you may want to discuss this in a revised version to be re-submitted.

Crude numbers and OR are now presented in Table 2. The results are very similar irrespectively of the analysis (univariate or multivariate). Changes in the Methods and Results units, as well as in Table 2, are in yellow.